Author’s response to reviews

Title: Exploring and analysing single cell multi-omics data with VDJView

Authors:
Jerome Samir (jsamir@kirby.unsw.edu.au)
Simone Rizzetto (simone.rizzetto@outlook.com)
Money Gupta (mgupta@kirby.unsw.edu.au)
Fabio Luciani (luciani@unsw.edu.au)

Version: 3 Date: 11 Feb 2020

Author’s response to reviews:

We have amended the manuscript according to the comments of the Assistant Editor. Below we respond to each comment:

1. Availability and requirements
   We have added the "Any restrictions to use by non-academics" section to the availability and requirements section.
   We have also updated the R version number (from 3.4.1 to 3.5.2) required to run VDJView, updating the "Other requirements" section on page 14, line 5, and in the implementation section on page 4, line 24.

2. Funding
   We have now provided the full name of the funders corresponding to “NHMRC” and “UNSW” in the Funding section of the manuscript.

3. Authors’ contributions
   We have corrected the discrepancy in the Authors’ contributions section and have now added Money Gupta's (MG) contributions.

4. Competing interests
   We have clarified the competing interests statement, adhering to required format, by stating that “The authors declare that they have no competing interests”.

5. Abbreviations
   We have provided a full list of alphabetically sorted abbreviations before the declarations section. All abbreviations are defined in the text at their first use.

6. Supplementary figures
   All supplementary figures are now referenced in the text. Supplementary figure 3 is referenced on page 7, line 14, and supplementary figures 5 - 7 are referenced on page 8, line 5.